Author's response to reviews

Title: Type V pit pattern: Impact of clinical experience and magnifying chromoendoscopy

Authors:

Taku Sakamoto (tasakamo@ncc.go.jp)
Takahisa Matsuda (tamatsud@ncc.go.jp)
Takeshi Nakajima (tnakajim@ncc.go.jp)
Yutaka Saito (ytsaito@ncc.go.jp)
Takahiro Fujii (tfclinic@khaki.plala.or.jp)

Version: 3
Date: 24 October 2013

Author's response to reviews:

The Editors

BMC gastroenterology

Dear Sir

I, along with my coauthors, would like to ask you to consider the enclosed manuscript entitled Type V pit pattern: the impact of clinical experience and magnifying chromoendoscopy for publication in the BMC gastroenterology as an Research Article.

The main conclusions of our study are that pit pattern analyses of colonic lesions with magnifying chromoendoscopy are a complimentary modality in estimating the invasion depth of early colorectal cancers, and that such diagnoses improve with the level of experience endoscopists.

We believe that the findings of this study are relevant to the scope of your journal and will be of interest to its readership.

This manuscript has not been published or presented elsewhere in part or in entirety, and is not under consideration by another journal. All study participants
provided informed consent, and the study design was approved by the appropriate ethics review boards. All the authors have approved the manuscript and agree with submission to your esteemed journal. There are no conflicts of interest to declare.

I look forward to hearing from you at your earliest convenience.

Sincerely,

Taku Sakamoto

Endoscopy Division

National Cancer Center Hospital

5-1-1 Tsukiji

Chuo-ku, Tokyo, 104-0045, Japan

Tel: +81-3-3542-2511

Fax: +81-3-3542-3815

E-mail: tasakamo@ncc.go.jp